# Assessing the change in soil water deficit characteristics from farmland to forestland on the Loess Plateau

Yongwang Zhang<sup>1</sup>, Mingyu Gao<sup>1</sup>, Ming Liu<sup>1</sup>, Kaibo Wang<sup>2</sup>, Jun Wang<sup>1</sup>, Changhai Liu<sup>3</sup>, and Zhouping Shangguan<sup>4</sup>

<sup>1</sup>Yan'an University <sup>2</sup>Chinese Academy of Science <sup>3</sup>Unknown <sup>4</sup>State Key Laboratory of Soil Erosion and Dryland Farming on the Loess Plateau

December 10, 2023

#### Abstract

Soil moisture is an important factor that affects terrestrial vegetation ecosystems and biological growth and development, and water deficit is one of the major environmental factors threatening vegetation restoration in the Loess Plateau. To determine the change in the soil water deficit characteristics from farmland to forestland on the Loess Plateau, in this study, we measured the soil water storage and deficit in abandoned grassland, shrubland, pioneer forestland and climax forestland along with vegetation succession. The results showed that the soil water content and the soil water storage with natural vegetation recovery from the abandoned grassland to shrubland and forestland showed a gradual declining trend, while on the contrary, the soil water deficit in July and September of the rainy season as well as that the vigorous growing period of plants was more serious than that in May and November, which are in the non-rainy season of the Chinese Loess Plateau. The vegetation types, soil texture and soil depth were the key factors affecting the soil water deficiency status in the process of vegetation succession. The results could have great potential in the sustainable development of forestry and provide a theoretical basis for effective water management and reasonable vegetation restoration in arid and semiarid loess regions.

#### Hosted file

981422\_0\_art\_1\_s5b16v.docx available at https://authorea.com/users/472676/articles/692226-assessing-the-change-in-soil-water-deficit-characteristics-from-farmland-to-forestland-on-the-loess-plateau

1	Assessing the change in soil water deficit characteristics from
2	farmland to forestland on the Loess Plateau
3	Yongwang Zhang <sup>a</sup> , Mingyu Gao <sup>a</sup> , Min Liu <sup>a</sup> , Kaibo Wang <sup>b</sup> , Jun Wang <sup>a</sup> , Changhai Liu <sup>a</sup> *,

4 Zhouping Shangguan<sup>c</sup>

<sup>a</sup>Shaanxi Key Laboratory of Chinese Jujube, Yan'an University, Yan'an, Shaanxi 716000,
China;

<sup>b</sup>Institute of Earth Environment, Chinese Academy of Sciences, Xi'an, Shaanxi, 710061,
China;

9 <sup>c</sup>Institute of Soil and Water Conservation, Chinese Academy of Sciences and Ministry of

10 Water Resources, Yangling, Shaanxi 712100, China

11 \*Corresponding author: yadxlch@yau.edu.cn

12 Abstract

Soil moisture is an important factor that affects terrestrial vegetation ecosystems and 13 biological growth and development, and water deficit is one of the major environmental 14 factors threatening vegetation restoration in the Loess Plateau. To determine the change in the 15 soil water deficit characteristics from farmland to forestland on the Loess Plateau, in this 16 study, we measured the soil water storage and deficit in abandoned grassland, shrubland, 17 pioneer forestland and climax forestland along with vegetation succession. The results 18 showed that the soil water content and the soil water storage with natural vegetation recovery 19 from the abandoned grassland to shrubland and forestland showed a gradual declining trend, 20 while on the contrary, the soil water deficit was increasing during succession, it was more 21 severe in the deep soil than in the shallow soil layers, and the soil water deficit in July and 22

September of the rainy season as well as that the vigorous growing period of plants was more 23 serious than that in May and November, which are in the non-rainy season of the Chinese 24 Loess Plateau. The vegetation types, soil texture and soil depth were the key factors affecting 25 the soil water deficiency status in the process of vegetation succession. The results could have 26 great potential in the sustainable development of forestry and provide a theoretical basis for 27 effective water management and reasonable vegetation restoration in arid and semiarid loess 28 regions. 29

#### **Plain Language Summary** 30

In this manuscript, we measured the soil water status under different land use types along with 31 the vegetation natural restoration from grassland to forestland in different seasons on the 32 Chinese Loess Plateau, and we find that the soil water storage was decreasing, and the soil 33 34 water deficit was increasing during the succession. Key words: Soil water deficit, Soil texture, Soil depth, Vegetation types, the Loess Plateau

36

37 **1. Introduction** 

According to current climate change models for the 21st century, an increase in average 38 temperature will likely occur in the near future, with more frequent extreme temperature and 39 drought episodes. Soil moisture is an important factor that affects terrestrial vegetation 40 ecosystems and biological growth and development [1], and water deficit is considered one of 41 the major environmental factors limiting plant growth, yield and survival in arid and semiarid 42 regions of the world, where plants are often exposed to drought conditions [2]. Located in the 43 northern part of central China, the Loess Plateau has an arid continental monsoon climate and 44 45 is one of the regions with the most serious soil erosion and the most fragile ecology in the world. In recent years, the government in China carried out the large-scale implementation of 46 returning farmland to forest and grass [3], which has made remarkable achievements; 47 48 however, it led to a lower soil moisture content, even though the dry layer phenomenon appeared because of the region with low rainfall and high evaporative conditions and because 49 of the unreasonable distribution of vegetation types in the process of vegetation restoration [4]; 50 51 additionally, along with vegetation restoration, the soil water deficiency phenomenon has become more severe [5-7]. This phenomenon is very unfavorable to the vegetation restoration 52 and ecological restoration of the Loess Plateau. 53

54 Due to the dry climate, scarce precipitation and strong evaporation on the Loess Plateau, 55 water resources have become the key factor in determining the structure and function of the 56 ecological system, and the soil moisture condition is the main factor affecting plant growth 57 and restoration and determining the type and structure of vegetation in ecological construction 58 [8]. Precipitation is the main source of soil moisture and water supply for plants in this area;

however, rainfall has a significant difference in seasonal distribution, which makes the 59 variation in soil moisture closely related to the change in seasonal precipitation [9] and 60 basically consistent with the change trend of seasonal precipitation. On the annual scale, 61 autumn and winter are the slow accumulation stages of soil moisture, late spring and early 62 summer are the intense consumption stages of soil moisture, summer is the fluctuation stage 63 of soil moisture, and soil dry layers occur widely on the Loess Plateau; some vertical soil 64 depths even appear as heavy dry layers [10]. The soil dry layers indicate that severe soil water 65 deficiency exists, and there are many reasons for the formation of dry soil layers, including 66 high plant density, less precipitation, global climate warming and drying and soil erosion; 67 however, the main reason in arid and semiarid areas is intense evaporation, which is attributed 68 to the dry climate. Therefore, scientifically taking advantage of land resources and managing 69 70 the ecological environment in loess hilly regions is necessary to fully understand the change and utilization of soil moisture under various land-use conditions and to study the spatial 71 change in soil water characteristics during the dry and wet seasons, which is of important 72 73 scientific value to improve the dry soil layers in the Loess Plateau. Many studies have been carried out to identify the factors affecting the soil water status. 74 Wang et al. analyzed soil desiccation in artificial forests and found that artificial forests could 75

vilize deep soil water to a threshold [11], resulting in a dry layer that causes the deep soil to

completely lose its ability to regulate the soil water status for plant growth. Zhu et al.

<sup>78</sup> indicated that soil water deficiency would be aggravated by intense evapotranspiration [12],

79 leading alfalfa to deepen its root distribution in arid and semiarid regions. As a result, soil

80 desiccation was much more serious than before in deep soil; nevertheless, the moisture

needed for vegetation evapotranspiration can be supplied by shallow groundwater [13], which 81 would decline with the increasing water demands of plants [14]. Both declining groundwater 82 and aggravating soil desiccation can mitigate the ability to resist drought and adversely affect 83 the soil water balance [15]. Soil water storage is the foundation of vegetation planting, and it 84 is consumed through transpiration and evaporation. The latter mainly expends the soil water 85 stored at the 0-60 cm soil depth [16], and root water uptake is the key mediator of plant 86 transpiration [17]. The soil water consumption may be higher where the roots are denser. Thus, 87 denser roots and sparser land coverage will increase transpiration and evaporation. With rising 88 global temperatures, evapotranspiration is increasing, which inevitably aggravates soil 89 desiccation in arid and semiarid regions [18]. A study in alfalfa grassland by Huang et al. 90 showed that the soil water deficit was severe in the 0~150 cm soil depth, and the deficit 91 92 degree was significantly different in alfalfa grasslands of different ages [19]. Soil water deficit has been serious during vegetation restoration on the Loess Plateau, which imposes great 93 obstacles to ecological restoration and vegetation construction on the Loess Plateau. Only by 94 95 fully understanding the characteristics of soil moisture and its deficiency on the Loess Plateau can we provide a theoretical basis for vegetation restoration. However, little is known about 96 the difference in the soil water deficit and the seasonal water storage characteristics in the 97 deep soil layers in different vegetation types during vegetation restoration. 98 In this study, we aimed to assess the spatial and temporal variation characteristics of soil 99 water storage and deficits in deep soil layers in different vegetation types in the process of 100 101 vegetation restoration; additionally, we aimed to determine how the soil water deficit changes in the dry and wet seasons during a year. Our findings could have great potential in the 102

103	sustainable development of forestry and provide a theoretical basis for effective water
104	management and reasonable vegetation restoration in arid and semiarid loess regions.
105	2. Materials and methods
106	2.1 Experimental site
107	The study area is located in the Lianjiabian forest farm of the Ziwuling forest area in
108	Heshui County, Gansu Province, the central part of the Chinese Loess Plateau, between 36°03'
109	and 36°05' N and 108°31' and 108°32' E, which belongs to the temperate continental monsoon
110	climate zone. The annual precipitation is 560~590 mm, mainly from July to September, and
111	the altitude is 1211~453 m. The region's soils are largely loessal having developed from
112	primitive or secondary loess parent materials, which are evenly distributed 50-130 m deep
113	above red earth consisting of calcareous cinnamon soil, and soil pH ranged from 7.92 to 8.31.
114	In this study, soil moisture was measured in the abandoned grassland (AG), secondary
115	Hippophae rhamnoides Linn. forestland (HF), Populus davidiana Dode. forestland (PF), and
116	Quercus liaotungensis Koidz. forestland (QF) to determine the soil water storage and water
117	deficit in May, July, September and November, respectively. The status of the vegetation types
118	is shown in Fig. 1.

# Figure 1

120 2	2.2 Sampling	and	measurements
-------	--------------	-----	--------------

121 From May to November 2018, soil samples were collected by soil drilling at the 122 beginning of each month. Standard wood was selected from each sample plot, and the soil 123 samples were drilled at a distance of 2/3 m from the tree canopy edge. The sampling soil 124 depths were 0~500 cm, with 10-cm intervals in the 0~100 cm layer and 20-cm intervals in the

125	soil layers below 100 cm. The soil moisture content was measured by the drying method.
126	Under the high temperature of 105°C, the soil was dried for 6~8 h until a constant weight was
127	reached.
128	The main vegetation in the study area is broad-leaved forest and coniferous forest, and
129	the top community is Quercus liaotungensis Koidz, with community coverage of 80%~95%.
130	In the Ziwuling forest area, historical records show that natural secondary forest began to
131	develop on abandoned farmland in the 1760s [20]. Therefore, this study mainly selected AG,
132	HF, PF and QF as the research objects. Five quadrats were set in each community, with five
133	replications, including 20 m×20 m for QF and PF, 5 m×5 m for HF, and 2 m×2 m for AG.
134	Community information is shown in Table 1.
135	Table 1
136	Soil water storage (SWS) refers to the absolute amount of water in the soil and was
137	calculated from Eq. (1) [21]:
138	$SWS_i = SWC_i \times BD_i \times h \times 10 \div 100 \tag{1}$
139	
	where $_i$ represents the sequence of soil layers, SWS <sub>i</sub> is the soil water storage (mm), BD <sub>i</sub> is the
140	where <i>i</i> represents the sequence of soil layers, <i>SWS<sub>i</sub></i> is the soil water storage (mm), <i>BD<sub>i</sub></i> is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and <i>SWC<sub>i</sub></i> is the soil water content (%).
140 141	where <i>i</i> represents the sequence of soil layers, $SWS_i$ is the soil water storage (mm), $BD_i$ is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and $SWC_i$ is the soil water content (%). The degree of relative soil water deficit was calculated using Eq. (2) [21]:
140 141 142	where <i>i</i> represents the sequence of soil layers, <i>SWS<sub>i</sub></i> is the soil water storage (mm), <i>BD<sub>i</sub></i> is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and <i>SWC<sub>i</sub></i> is the soil water content (%). The degree of relative soil water deficit was calculated using Eq. (2) [21]: $PD = \frac{\sum_{i=1}^{k} \frac{SWS_{ci} - SWS_{i}}{k}}{k}$ (2)
140 141 142 143	where <i>i</i> represents the sequence of soil layers, <i>SWS<sub>i</sub></i> is the soil water storage (mm), <i>BD<sub>i</sub></i> is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and <i>SWC<sub>i</sub></i> is the soil water content (%). The degree of relative soil water deficit was calculated using Eq. (2) [21]: $PD = \frac{\sum_{i=1}^{k} \frac{SWS_{ci} - SWS_{i}}{k}}{k}$ (2) where <i>PD</i> represents the soil water relative deficit index, <i>SWS<sub>i</sub></i> is the soil water storage (mm)
140 141 142 143 144	where <i>i</i> represents the sequence of soil layers, <i>SWS<sub>i</sub></i> is the soil water storage (mm), <i>BD<sub>i</sub></i> is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and <i>SWC<sub>i</sub></i> is the soil water content (%). The degree of relative soil water deficit was calculated using Eq. (2) [21]: $PD = \frac{\sum_{i=1}^{k} \frac{SWS_{ci} - SWS_{i}}{k}}{k}$ (2) where <i>PD</i> represents the soil water relative deficit index, <i>SWS<sub>i</sub></i> is the soil water storage (mm) in soil layer <i>i</i> of the sample lands, <i>SWS<sub>Ci</sub></i> represents the soil water storage (mm) in soil layer <i>i</i>
140 141 142 143 144 145	where <i>i</i> represents the sequence of soil layers, <i>SWSi</i> is the soil water storage (mm), <i>BDi</i> is the soil bulk density (g/cm <sup>3</sup> ), <i>h</i> is the soil depth (cm), and <i>SWCi</i> is the soil water content (%). The degree of relative soil water deficit was calculated using Eq. (2) [21]: $p_{D} = \frac{\sum_{i=1}^{k} \frac{SWS_{ci} - SWS_{i}}{k}}{k}$ (2) where <i>PD</i> represents the soil water relative deficit index, <i>SWSi</i> is the soil water storage (mm) in soil layer <i>i</i> of the sample lands, <i>SWSci</i> represents the soil water storage (mm) in soil layer <i>i</i> of the control lands, i.e., the AG, <i>SWSm</i> is the soil water storage (mm) corresponding to the

#### 147 **2.3 Statistical analysis**

Microsoft Excel 2010 (Microsoft Corporation, USA) was used to process experimental
data. Analysis of variance (ANOVA) was performed with the statistical software package
SPSS 20.0 (International Business Machines Corporation, USA). The least significant
difference (LSD) was at the 0.05 probability level. The graphics were created with Origin
software (OriginLab V8, USA).

153 **3. Results** 

# 154 **3.1 SWC in different vegetation types during vegetation restoration**

As shown in Fig. 2, the SWC first increased and then slowly decreased with increasing 155 soil depth in the HF, PF and QF, while it showed an unstable state in the AG. However, all the 156 forestlands exhibited an upward trend with increasing soil depth, and the SWC in the QF was 157 158 the lowest. This phenomenon was more obvious in the non-rainy season in May than in the other months, and the peak value of the SWC of the three forestlands in May was mostly 159 concentrated in the soil depth of 160 cm, ranging from 15~25%. The SWC was significantly 160 higher in July of the rainy season than in May of the non-rainy season, and the peak 161 distribution was different. The soil depth where the peak appeared in the HF was 140 cm and 162 was 100 cm and 80 cm in the PF and QF, respectively, ranging from 17~30%. In the rainy 163 season, the SWC in July, after passing the peak point, decreased obviously with increasing 164 soil depth. In the rainy season, the SWC content in September was the most stable and the 165 highest among the months from May to November. The SWC of the non-rainy season in 166 November decreased compared with that of the rainy season in September. In addition to the 167 QF, the average SWC of the 0~500 cm soil layer in rainy September and non-rainy November 168

was approximately 25%. When the soil depth was below 300 cm, the SWC in the AG was
significantly higher than that in the other three forestlands. Overall, from the perspective of
the four months above, the SWC in the AG was always at a high level, followed by that in the
HF, and then the PF, and the SWC in the QF was the lowest among the four vegetation types
during natural vegetation restoration.

174

## Figure 2

# 175 **3.2 SWS in different vegetation types during vegetation restoration**

As shown in Fig. 3, the variation trend of the SWS with soil depth in the three 176 forestlands is consistent with that of the SWC. With the increase in soil depth, the SWS in the 177 three forestlands first increased and then slowly declined, while it was unstable and increased 178 with soil depth in the AG. Among the four vegetation types during natural vegetation 179 180 restoration, the AG had the highest SWS, followed by that in the HF, and then the PF, and the SWS in the QF was the lowest, which is consistent with the SWC. It can be seen from the 181 SWS per meter depth in different vegetation types that the SWC in the AG increased with soil 182 183 depth in both the rainy season and the non-rainy season (Fig. 4), but this phenomenon was more significant in the non-rainy season than in the rainy season, and the SWS per meter 184 below 300 cm was the highest in the AG. In the five soil layers of 0~500 cm, with vegetation 185 recovery, there was a significant decline from shrubby (the HF) to arbor forest in the initial 186 stage (the PF) and then to arbor forest in the top stage (the QF); furthermore, the SWS in the 187 QF was the lowest except in the 0~100 cm soil layer in May, i.e., when it was not the rainy 188 season. Over 400 cm, the SWS in the HF in the rainy season and non-rainy season was the 189 highest, followed by that in the PF among the three forestlands. In general, the SWS during 190

191 the rainy season was higher than that during the non-rainy season.

192Figure 3193Figure 4

# 194 **3.3 The PD in different vegetation types during vegetation restoration**

The soil water relative deficit index (PD) varied significantly in different vegetation 195 types and in different seasons (Fig. 5). Regardless of rainy season or non-rainy season, the PD 196 in the QF was the highest, which meant that the soil water deficit phenomenon was most 197 obvious and the scarcity of water was most severe; PF ranked second, and the lowest PD 198 199 occurred in the HF. This result was inversely proportional to the SWC and SWS in the different vegetation types shown in Figs. 2 and 3. The PD in May and November of the 200 non-rainy season was higher than that in July and September of the rainy season, indicating 201 202 that the soil deficit in May and November was more serious than that in July and September of the rainy season, i.e., when the vigorous growth period of vegetation occurred. Compared 203 with the AG, the soil deficit in the QF was in a continuous state from May to November, but 204 205 the deficit situation changed during the whole growth period; from May to July, the deficit increased, and it slowed from July onwards. However, in the HF and PF, the soil water was 206 only slightly deficient in the soil layers below 300 cm in July and September, and the PD was 207 negative in the soil layers above 300 cm in all months, indicating there was no deficit. 208 209 Figure 5

Fig. 6 shows the difference in the PD per meter soil depth in different vegetation types from May to November. In the 0~100 cm soil layers in May of the rainy season, the PD in the PF was significantly higher than that in the HF and QF, and no deficit phenomenon was

213	observed in the QF; however, in the soil layers under 100 cm, the PD was significantly higher
214	in the QF than in the HF and PF. In the 100~300 cm soil layers, the PD was significantly
215	lower in the HF than in the PF, while in the 400~500 cm soil layers, the PD was significantly
216	higher in the HF than in the PF. In July and September of the rainy season and November of
217	the non-rainy season, the PD in the QF was significantly higher than that in the HF and PF in
218	all soil layers of 0~500 cm, while it was significantly lower in the HF than in the PF in the
219	100~300 cm soil layer except for September, and it was obviously higher in the HF than in the
220	PF in the 400~500 cm soil layer.
221	Figure 6
222	4. Discussion
223	4.1 Effects of vegetation on soil water deficit characteristics
224	Due to the difference in root density and depth, different vegetation types consume soil
225	water with different intensities and at different soil depths, resulting in differences in soil
226	water under forests with different vegetation types [22]. In our study, the SWC and SWS with
227	natural vegetation recovery from abandoned grass to shrubland and forestland showed a
228	gradual declining trend, possibly because the aboveground biomass increased gradually, the
229	foliage became more lush, and the tree body gradually enhanced its transpiration capability
230	precipitation interception and trunk runoff with vegetation restoration. In addition, the
231	increased root system underground, the augmented distribution, and the increased absorption
232	and consumption of soil water led to the overall decline of the SWC and SWS [10, 11].
233	Regarding the soil water deficiency phenomenon in this study, compared with the AG, the PD
234	in the QF, which was in the climax of the natural vegetation restoration process, was

significantly higher than that in the HR and PF, which were both in the intermediate stage of 235 natural vegetation restoration, and the soil water deficit was more severe in the deep soil than 236 in the shallow soil layers in both the rainy and the non-rainy season in all three vegetation 237 types. This result was due to the interaction between vegetation growth and soil moisture, the 238 QF consumed more water than the HR and PF, and due to its stronger canopy intercept and 239 lower amount of precipitation recharge, the ability of precipitation resources to ease the 240 degree of soil water deficit was greatly reduced and more unable to supplement the deeper 241 soil moisture, resulting in the deterioration of the soil water deficiency phenomenon in the 242 243 deep soil. This result was consistent with the results of previous studies [23, 24]. Gao et al. indicated that plants could consume a large amount of soil water storage due to the amount of 244 water consumed by the plants, which was higher than the precipitation balance in different 245 246 regions of the Loess Plateau, which would lead to the water storage deficit [22]. The research results of Chen et al. also showed that the soil depth where the negative compensation of 247 SWS occurred would reach approximately 300 cm [25]. The research results of Wang et al. 248 249 showed that the soil water deficiency phenomenon caused by the water consumption being greater than the recharge was widespread on the Loess Plateau [26], and the soil properties 250 and the meteorological factors were the determinants of the soil water deficiency of 251 determinants, which were not caused by bio-utilization. 252

### **4.2 Effects of soil properties on soil water deficit characteristics**

254 Soil moisture is affected not only by vegetation type but also by soil depth, particle 255 composition, infiltration and season. The research of Yang et al. showed that SWC decreased 256 significantly with vegetation restoration, with no significant difference in the soil water in the

soil surface in the different vegetation types [27], while it was significantly different in the 257 deep soil, and the soil water in the deep soil showed temporal stability to a certain degree, 258 which was consistent with the results of this study. The vertical SWC and SWS in the AG 259 showed a gradual increasing trend, which might have been because, as the soil layer deepened, 260 the grass root density decreased, and the water consumption correspondingly decreased. 261 However, the SWC and SWS in the HR, PF, and QF showed a trend that first increased and 262 then decreased, and the inflection point was at the 150 cm soil depth, which may be because 263 in the 150 cm soil layer, the root system gradually became dense, and its infiltration depth was 264 265 limited; thus, it cannot be replenished in a timely manner to the subsoil, causing the SWC in the lower layer to continue to decrease and even the dry soil layers appeared. In addition, the 266 SWS in the rainy season was significantly higher than that in the non-rainy season in this 267 268 study, but compared with the abandoned grassland, the shrubland and forestland had more evident soil water deficit phenomena in the rainy season than in the non-rainy season, which 269 suggested that the increased precipitation resources in the rainy season led the SWS in the 270 271 topsoil to increase to a certain extent and relieved the water deficit. Nevertheless, the 272 infiltration depth was not enough, and although precipitation was greater in the rainy season, which was also the vigorous growing period, the water consumption was greater than that in 273 the non-rainy season; thus, there was no significant improvement in the water deficit in deep 274 soil, and the soil water stored in the deep soil was lower in the rainy season than in the 275 non-rainy season. In addition, in the same soil layers, the SWC in the abandoned grassland 276 277 was higher than that in the shrubland and forestland, which agreed with the previous study of Yang et al. [28]. This result may be because of the high proportion of clay content in the 278

abandoned grassland, while the proportion of sand content was higher in shrubland and
forestland, and the clay can absorb more moisture than sand [29]. This process may be
another leading cause of the soil water deficit being higher in the shrubland and forestland
than in the abandoned grassland. Therefore, vegetation type, soil texture and soil depth were
the key factors affecting soil water deficit during vegetation succession.

#### 284 Conclusions

We found that the SWC in the HF, PF and QF first increased and then decreased slowly 285 with increasing soil depth, while it presented an unstable state in the AG. In general, it 286 showed an increasing trend with increasing soil depth. In both the rainy and the non-rainy 287 seasons, the SWC in the AG was the highest, followed by that in the HF, that in the PF, and 288 finally, that in the QF, which had the lowest SWC. From the non-rainy season to the rainy 289 290 season, the soil depth which the peak value of the SWC occurred in gradually increased from the abandoned grassland to the shrubland and forestland. Compared with the AG, the soil 291 water deficit in the QF was most serious, with a constant deficit from May to November, 292 293 followed by the PF and the HF. In addition, the soil water deficit in July and September of the rainy season and the vigorous growing period of plants were more serious than those in May 294 and November, which are in the non-rainy season in the Chinese Loess Plateau. The 295 vegetation types, soil texture and soil depth were the key factors affecting the soil water 296 deficiency status in the process of vegetation succession. This result can provide a reference 297 for the government to make policies on the rational utilization of water resources and the 298 sustainable development of vegetation restoration. 299

### 300 Acknowledgements

301	1 This study was sponsored by the National Natural Science Foundation of China							
302	(41807520, 32060297), the Innovation Capability Support Program of Shaanxi							
303	03 (2020KJXX-009).							
304	Ref	ferences						
305	1.	Legates, D.R., Mahmood, R., Levia, D., 2011. Soil Moisture: A central and unifying						
306		theme in physical geography. Progress in Physical Geography. 35(1), 65-86.						
307	2.	Yi, X.P., Zhang, Y.L., Yao, H.S., Luo, H.H., Gou, L., Chow, W.S., Zhang W.F., 2016.						
308		Rapid recovery of photosynthetic rate following soil water deficit and re-watering in						
309		cotton plants (Gossypium herbaceum L.) is related to the stability of the photosystems.						
310		Journal of Plant Physiology. 194, 23-24.						
311	3.	Shao, M.A., Jia, X.X., Wang, Y.Q., Zhu, Y.J., 2016. A review of studies on dried soil						
312		layers in the Loess Plateau. Advances in Earth Science. 31(1), 14-22.						
313	4.	Wang, L., Wei S.P., Wu, F.Q., 2009. Soil water environment and vegetation growth in the						
314		hilly and gully region of the Loess Plateau: a case study of Yangou Catchment. Acta						
315		Ecologica Sinica. 29(3), 1543-1553.						
316	5.	Bian, Y.Y., Song, N.P., Wang, X., Xiao, X.P., 2015. Soil water deficit under different						
317		land-use type in desert steppe. Journal of Soil and Water Conservation. 29(1), 201-206,						
318		213.						
319	6.	Liu, G., Niu, J.J., Zhu, W.X., Wang, L., 2015. Assessment on soil water deficit under						
320		different artificial forestland in loess hilly regions of northwestern Shanxi Province.						
321		Bulletin of Soil and Water Conservation. 35(5), 309-314.						
322	7.	Li, Y., Cohen, Y., Wallach, R., Cohen, S., Fuchs, M., 2004. On quantifying soil water						

- deficit of a partially wetted root zone by the response of canopy or leaf conductance.
- Agricultural Water Management. 65(1), 0-38.
- 8. Li, H.F., Wei, W., Chen, L.D., Guo, E.H., Huang, Y., 2013. Progress in the study of soil
- water balance under forest and grassland covers on the Loess Plateau. Research of Soil
- and Water Conservation. 20(1), 287-293.
- 328 9. Zhang, J.J., Wang, L., 2011. Surface soil moisture conditions in gully region of the Loess
  329 Plateau. Bulletin of Soil and Water Conservation. 31(1), 93-97.
- 10. Wang, Y.Q., Shao, M.A., Liu, Z.P., 2013. Vertical distribution and influencing factors of
- soil water content within 21-m profile on the Chinese Loess Plateau. Geoderma. 193,
  300-310.
- 11. Wang, Y., Shao, M.A., Liu, Z., Warrington, D.N., 2012. Investigation of factors
- controlling the regional-scale distribution of dried soil layers under forestland on the

Loess Plateau, China. Surveys in Geophysics. 33(2), 311-330.

- 12. Zhu, L., Zhang, H.L., Gao, X., Qi, Y.S., Xu, X., 2016. Seasonal patterns in water uptake
- for Medicago sativa, grown along an elevation gradient with shallow groundwater table
- in Yanchi County of Ningxia, Northwest China. J. Arid Land. 8, 1-14.
- 13. Ramos, T.B., Simionesei, L., Jauch, E., Almeida, C., Neves, R., 2017. Modelling soil
- water and maize growth dynamics influenced by shallow groundwater conditions in the
  Sorraia Valley region, Portugal. Agric. Water Manage. 185, 27-42.
- 14. Zheng, X., Zhu, J.J., Yan, Q.L., Song, L.N., 2012. Effects of land use changes on the
- 343 groundwater table and the decline of Pinus sylvestris var. mongolica plantations in
- southern Horqin Sandy Land, Northeast China. Agric. Water Manage. 109, 94-106.

345	15.	Wu, Y., Liu, T.X., Paredes, P., Duan, L.M., Pereira, L.S., 2015. Water use by a
346		groundwater dependent maize in a semi-arid region of Inner Mongolia:
347		evapotranspiration partitioning and capillary rise. Agric. Water Manage. 152, 222-232.
348	16.	Liu, X.P., He, Y.H., Zhang, T.H., Zhao, X.Y., Li, Y.Q., Zhang, L.M., Wei, S.L., Yun, J.Y.,
349		Yue, X.F., 2015. The response of infiltration depth, evaporation, and soil water
350		replenishment to rainfall in mobile dunes in the Horqin Sandy Land, Northern China.
351		Environ. Earth Sci. 73, 8699-8708.
352	17.	Zhang, W.T., Hu, G.Q., Dang, Y., Weindorf, D.C., Sheng, J.D., 2016. Afforestation and
353		the impacts on soil and water conservation at decadal and regional scales in Northwest
354		China. J. Arid Environ. 130, 98-104.
355	18.	Muluneh, A., Biazin, B., Stroosnijder, L., Bewket, W., Keesstra, S., 2015. Impact of
356		predicted changes in rainfall and atmospheric carbon dioxide on maize and wheat yields
357		in the Central Rift Valley of Ethiopia. Reg. Environ. Change. 15, 1105-1119.
358	19.	Huang, Z., Liu, Y., Cui, Z., Fang, Y., He, H.H., Liu, B.R., Wu, G.L., 2018. Soil water
359		storage deficit of alfalfa (Medicago sativa) grasslands along ages in arid area (China).
360		Field Crops Research. 221, 1-6.
361	20.	Wang, K.B., Shao, R.X., Shangguan, Z.P., 2010. Changes in species richness and
362		community productivity during succession on the Loess Plateau (China). Polish Journal
363		of Ecology 58, 549–558.
364	21.	Yang, L., Wei, W., Mo, B.R., Chen, L.D., 2011. Soil water deficit under different
365		artificial vegetation restoration in the semi-arid hilly region of the Loess Plateau. Acta

366 Ecologica Sinica. 31(11), 3060-3068.

367	22.	Gao. Y., F	Fan. J., Pe	eng. X.P.,	Wang, L	Mi. 1	M.X	2014.	Soil w	vater de	pletion a	and
						.,,			~ ~ ~ ~ ~		p	

- infiltration under the typical vegetation in the water-wind erosion crisscross region. Acta
   Ecologica Sinica. 34(23), 7038-7046.
- 370 23. Guo, Z.S., 2009. Using depth of soil water and water consumption by little leaf peashrub
- in the semiarid area of loess hilly region. Bulletin of Soil and Water Conservation. 29(5),69-72.
- 24. Cheng J.M., Wan, H.E., Wang, J., Yong, S.P., 2003. Soil water regulation of the natural
  grassland of semi-arid loess hilly region. Acta Agrestia Sinica. 11(4), 296-300.
- 25. Chen, H.S., Shao, M.A., Wang, K.L., 2005. Water cycling characteristics of grassland
  and bare land soils on Loess Plateau. Chinese Journal of Applied Ecology. 16(10),
  1853-1857.
- 26. Wang, L., Shao, M.A., Wang, Q.J., Jia, Z.K., Li, J., 2004. Review of research on soil
- desiccation in the Loess Plateau. Transactions of the Chinese Society of Agricultural
  Engineering. 20(5), 27-31.
- Yang, L., Wei, W., Chen, L.D., Chen, W., Wang, J., 2014. Response of temporal variation
  of soil moisture to vegetation restoration in semi-arid Loess Plateau, China. Catena. 115,
  123-133.
- Yang, L., Wei, W., Chen, L.D., Mo, B.R., 2012. Response of deep soil moisture to land
  use and afforestation in the semi-arid Loess Plateau, China. Journal of Hydrology. 475,
  111-122.
- Satyanaga, A., Rahardjo, H., Leong, E.C., Wang, J.Y., 2013. Water characteristic curve of
  soil with bimodal grain-size distribution. Computers & Geotechnics. 48(4), 51-61.

#### **Table and Figure captions:**

Table 1 Geographical information and vegetation types in the sample plots during the vegetation restoration. Note: AG represents the abandoned grassland, HF stands for the shrubland, PF represents the pioneer forestland and QF stands for the climax forestland. And the restoration ages after cropland abandonment were 10, 50, 100 and 160, respectively.

Fig. 1. The status of the vegetation types in the study area.

Fig. 2. Vertical SWC change at different time in different vegetation types during the restoration.

Fig. 3. Vertical SWS change at different time in different vegetation types during therestoration.

Fig. 4. The difference of the SWS in the per meter soil layers at different time in different vegetation types during the restoration. Values are in the form of Mean  $\pm$  SE and the sample size n=5. Different uppercases above the bars mean significant differences in the different soil layers in the same vegetation types, and different lowercases above the bars mean significant differences in the same soil layers in the different vegetation types (P < 0.05).

Fig. 5. Vertical PD change at different time in different vegetation types during therestoration.

Fig. 6. The difference of the PD in the per meter soil layers at different time in different vegetation types during the restoration. Values are in the form of Mean  $\pm$  SE and the sample size n=5. Different lowercases above the bars mean significant differences in the same soil layers in the different vegetation types (P < 0.05)

Vegetation	Latitude	Longitude	Altitude	<b>A</b>	Slope	Coverage	Main along an aire
types	(N)	(E)	(m)	Aspect	(°)	(%)	Main plant species
AG	36°5′4″	108°31'37″	1348	NE	14	85	Lespedeza bicolorr
HF	36°4′14″	108°32′1″	1354	NE	18	90	H. rhamnoides Linn.
PF	36°2′16″	108°31'32″	1450	NE	12	85	P. davidiana, Dode.
QF	36°2′57″	108°32′13″	1449	NE	18	95	Q. liaotungensis Koidz.

Table 1 Geographical information and vegetation types in the sample plots during thevegetation restoration.

414 Note: AG represents the abandoned grassland, HF stands for the shrubland, PF represents the

415 pioneer forestland and QF stands for the climax forestland. And the restoration ages after

416 cropland abandonment were 10 a, 50 a, 100 a and 160 a, respectively.





Fig. 1. The status of the vegetation types in the study area.



421 Fig. 2. Vertical SWC change at different time in different vegetation types during the

422 restoration.



Fig. 3. Vertical SWS change at different time in different vegetation types during therestoration.



Fig. 4. The difference of the SWS in the per meter soil layers at different time in different vegetation types during the restoration. Values are in the form of Mean  $\pm$  SE and the sample size n=5. Different uppercases above the bars mean significant differences in the different soil layers in the same vegetation types, and different lowercases above the bars mean significant differences in the same soil layers in the different vegetation types (*P* < 0.05).



Fig. 5. Vertical PD change at different time in different vegetation types during the restoration.



Fig. 6. The difference of the PD in the per meter soil layers at different time in different vegetation types during the restoration. Values are in the form of Mean  $\pm$  SE and the sample size n=5. Different lowercases above the bars mean significant differences in the same soil layers in the different vegetation types (P < 0.05).